Author’s response to reviews

Title: Using Implementation Facilitation to Implement Primary Care Mental Health Integration via Clinical Video Telehealth in Rural Clinics: Protocol for a Hybrid Type 2 Cluster Randomized Stepped Wedge Design

Authors:

Richard Owen (Richard.Owen2@va.gov)
Eva Woodward (Eva.Woodward2@va.gov)
Karen Drummond (Karen.Drummond@va.gov)
Tisha Deen (Tisha.Deen@va.gov)
Karen Oliver (Karen.Oliver2@va.gov)
Nancy Petersen (Petersen.NancyJ@va.gov)
Scott Meit (Scott.Meit@va.gov)
John Fortney (John.Fortney@va.gov)
JoAnn Kirchner (JoAnn.Kirchner@va.gov)

Version: 2 Date: 22 Feb 2019

Author’s response to reviews:

Please see the uploaded response letter addressing the reviewer and editor comments. Thank you.